Author's response to reviews

Title: Disaster resilience in tertiary hospitals: A cross-sectional survey in Shandong Province, China

Authors:

Shuang Zhong (amigo-008@163.com)
Xiang-Yu Hou (x.hou@qut.edu.au)
Michele Clark (mi.clark@qut.edu.au)
Yu-Li Zang (yulizh@sdu.edu.cn)
Ling-Zhong Xu (lz xu@sdu.edu.cn)
Gerry FitzGerald (gj.fitzgerald@qut.edu.au)

Version: 2
Date: 22 October 2013

Author's response to reviews: see over
Dear Editor,

**Title:** Disaster resilience in tertiary hospitals: A cross-sectional survey in Shandong Province, China

**Changes made:**

1. Include email addresses of all authors in the title page;
2. Include a ‘competing interests’ section in the manuscript;
3. Include ‘authors’ contributions’ section in the manuscript

Sincerely yours,

Shuang Zhong (On behalf of the authors)